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TORI-01 – Giordano Bruno

Status: working

Problems: Low rate (10-30 Hz) + high currents
→ next year clima installation + washing the 

chambers

HW needed: if possible since we have a 
Hytech GPS….



  

TORI-02 – Alessandro Volta

Status: switched off because of trigger card

Problems:

→ trigger card
→ possible problem on one TDC 64 bit (to be 

checked)

IMPORTANT for NEXT RUN for triple 
coincidences with TORI-03-04



  

TORI-03 – Galileo Ferraris

Status: working

No Problems

Clima installed



  

TORI-04 – Massimo D’Azeglio

Status: working

No Problems

Clima installed



  

TORI-05 – Gobetti Segrè / Dip Fisica

Status: to be installed in September

TDCs/V1718/crate/structure/LV-HV 
cabling/gas pipes/gas bottles/PC

purchased by INFN Turin

Needed:

Data cables, trigger card + GPS

Next week the online datasheet will be filled
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